Author’s response to reviews

Title: Overexpression of CDC2/CyclinB1 in gliomas, and CDC2 depletion inhibits proliferation of human glioma cells in vitro and in vivo

Authors:

- Chen Hua (cutv1220@yahoo.com.cn)
- Huang Qiang (hq1936@yahoo.com.cn)
- Dong Jun (didongjun@163.com)
- Zhai De zhong (chuliang_1980@yahoo.com.cn)
- Wang Ai dong (eagle.wangad@yahoo.com.cn)
- Lan Qing (weizilong2007@163.com)

Version: 3 Date: 30 July 2007

Author’s response to reviews: see over
Dear editors:

Here we submit our paper of “Overexpression of CDC2/CyclinB1 in gliomas, and CDC2 depletion inhibits proliferation of human glioma cells in vitro and in vivo” to BMC Cancer. The manuscript ID is 1551981497153521. Before doing this, we give our statements as follow: 1) All authors have seen and approved the manuscript; 2) There is no conflict of this paper exists; 3) Neither the submitted paper nor any similar paper, in whole or in part, has been or will be submitted to or published in any other printed or digital publication.

Our manuscript firstly found that the positive expression rates of CDC2/CyclinB1 were positively correlated with the pathologic grades of human gliomas. Downregulation of CDC2 expression could inhibit the proliferation, induce G2/M arrest, and trigger apoptosis in human gliomas in vitro and in vivo. Therefore, CDC2 could become a potential target on gene therapy of human gliomas.

We are appreciated of your reviewing and look forward to hearing from you soon.

Yours sincerely

Chen Hua, Huang Qiang, Dong Jun, Zhai De-zhong, Wang Ai-dong, Lan Qing